

Citation: McIntyre-Nolan S, Kumar V, Mark-Carew M, Kumar K, Nightingale ES, Dalla Libera Marchiori G, et al. (2023) Comparison of collection methods for *Phlebotomus argentipes* sand flies to use in a molecular xenomonitoring system for the surveillance of visceral leishmaniasis. PLoS Negl Trop Dis 17(9): e0011200. https://doi.org/10.1371/journal.pntd.0011200

Editor: Claudia Ida Brodskyn, Centro de Pesquisa Gonçalo Moniz-FIOCRUZ/BA, BRAZIL

Received: February 27, 2023

Accepted: August 14, 2023

Published: September 1, 2023

Copyright: © 2023 McIntyre-Nolan et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data are available at: https://github.com/esnightingale/vl-collectionmethods-xm.

Funding: This study was supported by the Bill and Melinda Gates Foundation (<u>https://www.</u> gatesfoundation.org/) through the SPEAK India consortium [OPP1183986] (SMN, VK, MM-C, KK, ESN, GDLM, MER, MK, SC, GFM, PD, MMC). The views, opinions, assumptions or any other RESEARCH ARTICLE

Comparison of collection methods for *Phlebotomus argentipes* sand flies to use in a molecular xenomonitoring system for the surveillance of visceral leishmaniasis

Shannon McIntyre-Nolan¹, Vijay Kumar², Miguella Mark-Carew¹, Kundan Kumar², Emily S. Nightingale³, Giorgia Dalla Libera Marchiori¹, Matthew E. Rogers¹, Mojca Kristan¹, Susana Campino⁴, Graham F. Medley³, Pradeep Das^{5‡}, Mary M. Cameron^{1‡*}

 Department of Disease Control, London School of Hygiene & Tropical Medicine, London, United Kingdom,
Rajendra Memorial Research Institute of Medical Sciences, Patna, India, 3 Department of Global Health and Development, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 Department of Infection Biology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 5 Department of Molecular Parasitology, National Institute of Cholera and Enteric Diseases, Kolkata, India

So These authors contributed equally to this work.

‡ PD and MMC also contributed equally to this work.

* mary.cameron@lshtm.ac.uk

Abstract

Background

The kala-azar elimination programme has resulted in a significant reduction in visceral leishmaniasis (VL) cases across the Indian Subcontinent. To detect any resurgence of transmission, a sensitive cost-effective surveillance system is required. Molecular xenomonitoring (MX), detection of pathogen DNA/RNA in vectors, provides a proxy of human infection in the lymphatic filariasis elimination programme. To determine whether MX can be used for VL surveillance in a low transmission setting, large numbers of the sand fly vector *Phlebotomus argentipes* are required. This study will determine the best method for capturing *P. argentipes* females for MX.

Methodology/Principal findings

The field study was performed in two programmatic and two non-programmatic villages in Bihar, India. A total of 48 households (12/village) were recruited. Centers for Disease Control and Prevention light traps (CDC-LTs) were compared with Improved Prokopack (PKP) and mechanical vacuum aspirators (MVA) using standardised methods. Four 12x12 Latin squares, 576 collections, were attempted (12/house, 144/village,192/method). Molecular analyses of collections were conducted to confirm identification of *P. argentipes* and to detect human and *Leishmania* DNA. Operational factors, such as time burden, acceptance to householders and RNA preservation, were also considered. A total of 562 collections (97.7%) were completed with 6,809 sand flies captured. Females comprised 49.0% of captures, of which 1,934 (57.9%) were identified as *P. argentipes*. CDC-LTs collected 4.04 times more *P. argentipes* females than MVA and 3.62 times more than PKP (p<0.0001 for

information set out in this article are soley those of the authors and should not be attributed to the funders or any person connected to the funders. The funders had no role in the study design, data collection, decision to publish, or preparation of the manuscript.

Competing interests: The authors declare that they have no competing interests.

each). Of 21,735 mosquitoes in the same collections, no significant differences between collection methods were observed. CDC-LTs took less time to install and collect than to perform aspirations and their greater yield compensated for increased sorting time. No significant differences in *Leishmania* RNA detection and quantitation between methods were observed in experimentally infected sand flies maintained in conditions simulating field conditions. CDC-LTs were favoured by householders.

Conclusions/Significance

CDC-LTs are the most useful collection tool of those tested for MX surveillance since they collected higher numbers of *P. argentipes* females without compromising mosquito captures or the preservation of RNA. However, capture rates are still low.

Author summary

Molecular xenomonitoring, screening insects for pathogen DNA/RNA, may be used for surveillance of diseases transmitted by insects. Since the proportion of insects infected with pathogens is very low in areas targeted for disease elimination, large numbers of females need to be screened. We compared three different methods for collecting Phlebotomus argentipes sand fly females, the vector of parasites causing the disease visceral leishmaniasis in the Indian subcontinent, to determine which collected the largest number of females. Other factors that may also influence selection of a particular method of collection by a disease control programme, such as the time it takes to collect and sort samples, the acceptance of householders for a particular collection method and whether RNA degradation in insect samples differed between collection methods, were also considered. Centers for Disease Control and Prevention light traps (CDC-LTs) proved to be more useful than two types of aspiration methods for collecting higher numbers of sand fly females and RNA preservation was retained. Furthermore, they took less time to install than to perform aspirations and were favoured by householders. Therefore, CDC-LTs were considered to be the most suitable collection method for molecular xenomonitoring of sand flies in India.

Introduction

Visceral Leishmaniasis (VL) Epidemiology and Transmission in India

Across the Indian Subcontinent (ISC) visceral leishmaniasis (VL), also known as kala-azar, is an anthroponotic disease caused by the parasite *Leishmania donovani* and vectored by only one incriminated sand fly species, *Phlebotomus argentipes* [1]. In 2005, it was estimated that there were 500,000 cases of VL in the ISC, and around 147 million people living in Bangladesh, India and Nepal were at risk of the disease [2]. Given its epidemiology in the region, with humans being the only reservoir host and *P. argentipes* the only vector, together with newer tools for diagnosis (rk 39 dipstick test), safe treatment (miltefosine) and vector control using indoor residual spraying (IRS), the prospect of elimination of VL as a public health problem became a feasible goal [2]. Consequently, a memorandum of understanding was signed by the three regional members at the WHO Headquarters in Geneva, Switzerland to eliminate VL as a public health problem with an agreed target to reduce incidence to below 1 case/10,000

population at the country's appropriate administration level, i.e. upazila in Bangladesh, block in India and sub-district in Nepal, by 2015 [2].

Since the implementation of the elimination strategy, consisting of rapid case detection, treatment of VL cases and integrated vector management, and the accelerated plan for elimination practiced by the National Center for Vector Borne Disease Control, formerly the National Vector Borne Disease Control Programme, in India [3], the geographical distribution of the burden of VL has changed dramatically [4]. The number of VL cases in Bangladesh, India, and Nepal reported to the WHO declined sharply, from 50,091 of 62,715 new autochthonous VL cases reported globally in 2006 (80% of the global burden) to 4,692 of 22,691 new autochthonous VL cases reported from 60 of 78 countries endemic for VL in 2018 (an unprecedented low of 27% of the global burden) [4]. Bangladesh and Nepal have reached the elimination target in all endemic areas, while India has reached the target in 714/751 (95%) endemic blocks (i.e. those deemed to have on-going transmission). In 2019, all 37 blocks that were above target were located in two of four VL endemic states: Bihar and Jharkhand (21 and 16 blocks, respectively) [5].

This progress is welcomed, but it brings enormous challenges to the Regional Kala-azar Elimination Programme since, as VL cases become scarce, it is crucial to have sensitive and cost-effective surveillance systems in place to determine whether elimination targets have been met and sustain control to prevent resurgence. Molecular xenomonitoring (MX) is a vector-based pathogen surveillance system, detecting pathogen DNA/RNA in a vector as a proxy of human infection, that may serve as a useful alternative to monitoring human infection and residual transmission in this low-transmission context [6].

Molecular Xenomonitoring (MX)

Much of the previous literature on MX concerns its use in lymphatic filariasis (LF) elimination programmes and, through several decades of surveillance during mass drug administration (MDA) programmes measuring human and entomological parameters concurrently, interruption of transmission is indicated if: the circulating filarial antigen is <2%, microfilarial rate is <0.5%, antibody rates in first-year primary school children is <2% and/or if Wuchereria bancrofti DNA infection rates in mosquito pools are < 0.25%, and these thresholds are considered as endpoints of transmission for LF [7,8]. Establishing the appropriate sample size for a given MX system depends on its goal. For elimination, endpoints of transmission are used to determine if elimination is nearing, has been achieved, or has been sustained [6]. Sample size calculations depend on estimates from existing literature on vector abundance and infection prevalence (ideally from the same, nearby, or characteristically similar sites). Where the infection prevalence in humans is high, it can be assumed that infection in the vector will also be high, so in areas where elimination activities are underway, and the human infection prevalence will decrease, the infection rates will be lower in vectors and more vectors will be required to detect parasite DNA in pools [6]. Other factors that need to be considered, as shown for LF elimination surveillance, are geographical settings, vector efficiency and vector/parasite pairings as the numbers of mosquitoes required to obtain accurate filarial prevalence rates varies and may be as high as 22,000 Aedes, 5,000-7,500 *Culex quinquefasciatus* or 2,500 *Anopheles* females [9].

Consequently, several studies were performed to obtain the large numbers of mosquitoes required and to gain an understanding of their behaviour since this influences the ideal tool and method of collection [6]. In general, for *Cx. quinquefasciatus*, Centers for Disease Control (CDC) and Prevention gravid traps (GTs) baited with odorous infusions, exploiting oviposition behaviour, are placed outside of houses and are the preferred method for collecting gravid adult females. However, a comparison of collection methods in Brazil found that large battery-

powered aspirators, exploiting host-seeking behaviour, collected 38 times more blood-fed and 5 times more gravid *Cx. quinquefasciatus* than CDC light traps (LTs) placed inside houses [10]. Furthermore, residents preferred aspirators over fixed battery or AC powered traps (e.g., CDC-LTs) due to lower risk of battery theft, power cuts, and the nuisance of light and noise in the bedroom at night [10]. Similarly, for arbovirus surveillance, common methods for collect-ing mosquitoes have known advantages and disadvantages [11]. However, the optimal collection method to use for *P. argentipes*, to develop an MX system for *L. donovani* DNA surveillance, has not been explored.

Rationale for MX system for VL surveillance in India

A better understanding of the transmission dynamics of VL, in particular of how rates of infection in humans and sand flies vary as functions of each other, is required to guide VL elimination efforts and ensure sustained elimination in the ISC. By collecting contemporary entomological and human data in the same geographical locations with contrasting levels of endemicity, similar to the approaches used in the aforementioned LF studies [6], infection prevalence of VL in humans can be quantified and related to L. donovani DNA prevalence in P. argentipes to generate precise epidemiological models and, in so doing, the thresholds for endpoints of transmission for VL can be determined. Furthermore, once the relationship between prevalence in humans and prevalence in vectors is identified, L. donovani DNA prevalence rates in vectors can be used as a proxy of human infection in a MX surveillance system that can be used rather than, or complementary to, human case surveillance. For example, the entomological inoculation rate (EIR), a product of the human bloodmeal index (HBI) and the proportion of vectors that are infective, is a useful proxy for human infection in malaria control programmes to evaluate the potential efficacy of intervention tools, but is unknown for VL [12]. In order to calculate the EIR for VL in the ISC, a suitable collection method to obtain large numbers of *P. argentipes* females is required.

Systematic longitudinal sampling of *P. argentipes* populations to monitor the impact of IRS on their abundance, HBI, and the presence of *L. donovani* infection is not performed routinely in India, Bangladesh or Nepal. Nevertheless, several research studies have captured *P. argentipes* for different purposes, and the most common methods use either mouth aspirations or CDC-LTs placed in human dwellings or in cattle sheds [13]. Specific details concerning trap placement and measures taken to standardize collections, which is a particularly important concern when mouth aspirations are used, are often missing from surveys making comparisons between them difficult. Furthermore, collections of blood-fed females are often limited, which adds to the potential bias arising through sampling procedures and can lead to data misinterpretation [14]. More recently, studies in Bihar have improved the standardization of collections using CDC-LTs and found that *P. argentipes* feed preferentially on humans and, a higher proportion of human-fed *P. argentipes* are found in cattle enclosures (55%) compared with houses (31%) [15,16].

The first study to examine natural infection rates of *P. argentipes* with *L. donovani* in Bihar collected a total of 14,585 sand flies using CDC miniature light traps and mouth aspirators in the Muzaffarpur district [17]. Of these, a subset of 449 *P. argentipes* females were divided into pools for molecular detection of the 18S rRNA gene using PCR, but the overall prevalence of infection in *P. argentipes* for *L. donovani* DNA was over estimated as 4.90–17.37% because the number of individual sand flies that may be positive in a single pool was not taken into account [17]. Subsequently, more reliable estimates were obtained using individual sand flies and infection rates of *P. argentipes* with *L. donovani* DNA were found to vary according to season: 1.0% (4/384) in summer, 0.9% (5/591) in the rainy season and 2.8% (12/422) in winter [18].

However, this study was performed over a single year, and morphological identifications were not confirmed using molecular methods, so further work is required to examine seasonal trends. Since then, studies have collected *P. argentipes* using CDC-LTs and/or mouth aspirators [16,19] but no natural infection rates of *P. argentipes* for *L. donovani* DNA were published in the last seven years prior to the commencement of the present study.

The primary aim of the present study is to compare collection methods to determine which method collects the largest number of *P. argentipes* females for use in subsequent epidemiological studies where the EIR and endpoints for transmission will be calculated. CDC gravid traps (CDC-GTs), baited with water rather than organic infusions for indoor use, were a method considered for inclusion because of their widescale deployment in LF surveillance, but they were not selected in the present trial since it was previously shown that collections of sand flies (including live as well as dead females of different physiological stages), were higher in CDC-LTs than in CDC-GTs in Odisha, India [20]. Therefore, CDC-LTs, often considered a 'gold standard' for *P. argentipes* [13], will be compared with two different types of battery-operated aspirators that were shown to collect more mosquitoes resting indoors than CDC-LTs or CDC backpack aspirators in earlier trials [10,21]. Although *P. argentipes* is the primary species of interest, incidental captures of mosquito females will also be analysed. In addition to entomological indicators, including human bloodmeal analysis and *L. donovani* infection rates, operational factors relating to the time burden of different collection methods, and their acceptance to householders, will be examined.

Another important consideration when selecting an appropriate collection technique for MX is whether pathogen RNA is preserved in the vector during collection and subsequent transportation prior to storage at -80°C. RNA detection is required for screening vectors for the infective stages of parasites, the metacyclic stages in the case of *Leishmania*, to measure EIR. To preserve the integrity of samples, a cold chain is required during transport of samples to the laboratory where they can be placed in RNA preservative media and frozen [11,22]. However, it is possible that the different conditions experienced by sand flies between collection methods, where sand flies are placed in a cool bag immediately after aspiration versus being trapped in a CDC-LT for possibly 12 or more hours before being placed in a cool bag, may compromise RNA preservation which will be addressed through a laboratory experiment simulating field conditions.

Methods

Ethics statement

The study was approved by the Ethics committee of the Government of India Health Ministry's Screening Committee (Ref: 2017–4126), Rajendra Memorial Research Institute of Medical Sciences (Ref: 39/RMRI/EC/2017) and the London School of Hygiene & Tropical Medicine (Ref: 1463). Members of each household were provided with an information sheet in Hindi explaining the purpose of the study, their involvement, and any potential adverse outcomes prior to commencement of sampling at each site. Information sheets were read to participants if they identified themselves as illiterate. Participants met with the investigators to answer any questions about the study. The household head or a nominated proxy provided written or oral informed consent for recruitment to the study. Signed consent forms and audio consent recordings were stored securely to protect participant confidentiality.

Study sites

The field study was performed in Bihar, the state in India with the highest incidence of VL. Two villages each in Nalanda (Ruchanpura, also known as Kosiawan, in Ekangarsarai block



Fig 1. Location of study villages and Rajendra Memorial Research Institute of Medical Sciences in Bihar, India. Base layer available at: World Topo Map (MapServer) (arcgisonline.com). Map generated using Epi Info 7 software.

https://doi.org/10.1371/journal.pntd.0011200.g001

and Dharampur in Thartari block) and Saran (Bishambharpur and Rampur Jagdish, both in Dariapur block) districts were selected for sand fly collection (Fig 1). In 2018, population sizes in each of the four villages/wards where the study was performed were 643 (Ward 4), 1209 (Ward 2), 3344 and 2078, respectively. Nalanda is located south of the Ganges River (25° 12′ 0″ N, 85° 31′ 12″ E); the Ganges River forms the southern border of Saran (25° 55′ 0″ N, 84° 45′ 0″ E). Travel distance to the villages from Rajendra Memorial Research Institute of Medical Sciences (RMRIMS), where the field team were based, ranged from 41–50 kilometres (per Google Maps) and travel time was estimated to be between 1.5 to 2.5 hours depending on traffic conditions.

Nalanda villages were defined as non-programmatic villages, i.e. at the time of sampling they met and sustained the Government of India's elimination target of <1 VL case reported/ 10,000 population at the block level over a three-year period, and no IRS or other vector control activities occurred in the five years preceding this study. Saran villages were defined as programmatic villages, meaning that they were above the elimination target at the time of sampling and should receive IRS twice during the calendar year.

Collection Method Comparison (CMC)

Sample size calculation. Prior to the start of the study, a sample size calculation using data collected during a previous trial conducted in Bihar, where a mean of 1.9776 *P. argentipes*

females, with a standard deviation of 3.2512, were collected with CDC-LTs per trap night [23], determined that 166 trap nights/collection method was sufficient to detect a difference of one sand fly in the mean number of *P. argentipes* females collected by each of the three methods (where $\alpha = 0.05$; $\beta = 0.20$). To determine the best practice for sampling female *P. argentipes* sand flies, CDC-LTs (John Hock) were compared with Improved Prokopack aspirators (PKP) (John Hock) and mechanical vacuum aspirators (MVA) (Horst Armadilhas).

Design. Households were selected randomly and householders were invited to take part in the study. Householders were considered eligible for inclusion if: 1) informed consent was obtained, 2) sand flies were found resting in their homes during a pre-study vector survey, and 3) they were willing to provide access to their homes for insect collections throughout the proposed study period. Households were excluded if they did not meet the inclusion criteria. The final selection of households included human-only and combined human-cattle dwellings. A total of 48 households (12 from each of the four villages) were recruited for regular indoor sampling four days a week over 12 weeks from 25 June to 14 September 2018. Using a Latin Square design balanced for carryover effects, the three collection methods were rotated through each household with four 12x12 Latin Squares (rounds) over the study period. A total of 576 collection events were expected (288 events per district, 144 events per village, 192 collection events per collection method, and 12 collection events per household). Replacement was needed for three households over the course of the study either because homes were no longer accessible or because householders no longer wished to participate.

Householder acceptance. To understand acceptance of and preferences for the three collections methods, all 48 households were surveyed to determine their preferred collection method, the rationale for their choice, and other considerations for the three methods. The investigators asked householders the following questions:

- 1. Do you prefer collections performed by: CDC light traps, mechanical vacuum aspirators, Prokopack aspirators or have no preference?
- 2. What is your reason for the answer given to question 1?
- 3. Do you have any additional comments/complaints relating to any of the collection methods?

CDC-LT Protocol. When allocated, a CDC-LT was installed in each of participating households and turned on between 18.00 hours and 06.00 hours based on the day of treatment according to the Latin square design. Batteries (6V 12 amp) were charged at RMRIMS prior to installation. CDC-LTs were hung in the room where most household residents slept at a distance 15cm from the wall of the house with the base of the collection pot 5-10cm from the floor. A member of the field team connected one receptor of the CDC-LT to the appropriate terminal of the battery and instructed a household member on how to connect the second receptor between 18.00 and 19.00 on the installation day. Householders were contacted via mobile phone near this time to ensure that traps were turned on for the expected duration. The field team returned the next morning to collect the CDC-LTs.

PKP and MVA Protocols. At the time of recruitment, the internal area of the room where PKP and MVA aspirations were to be performed was measured. The amount of time spent aspirating each room was calculated based on this room-specific measurement and an aspiration speed of approximately of $1m^2$ per 30 seconds. Two PKP (12V 12 amp) and two MVA (12 V 5 amp) batteries were charged overnight prior to use the following morning. Aspirations began between 07:00 and 08:00 and continued until all houses assigned to these collection methods on the day were completed.

Two field workers were needed to perform PKP and MVA aspirations: one to operate the PKP or MVA and another to operate a timer and remove objects obstructing the aspiration path. Aspirations always commenced on the wall immediately to the left of the doorway, on entering the room, and were continued clockwise around the room. Space below furniture, such as beds and tables, was also aspirated. In instances where the operator completed all walls before the allocated time had expired, a second round commenced until the calculated aspiration time was reached. Walls were aspirated to a height of up to approximately seven feet from the floor or to the ceiling, whichever was lower. Aspiration occurred left to right across the wall, at a distance of approximately 1cm from the surface, and gradually shifted downwards covering the entire area to the bottom of the wall. Contents of collection tubes were observed after aspiration for the presence of spiders, which if found were killed due to the risk of predation of sand flies.

Post-collection storage, transport, and processing. Specimens from all three collection methods were transported in holding containers placed in insulated coolers, containing ice packs, to RMRIMS. Due to the limited number of capture bags available for the MVAs, and collection containers for the PKP, aspirations were transferred to universal tubes fitted with fine mesh and secured by a screw cap with a hole in it for transfer of specimens using a small, handheld battery-operated aspirator. Holes were plugged with moist cotton wool to retain humidity and were secured with autoclave tape immediately after transfer for transport and storage. Universal tubes and CDC-LT collection pots were all stored at -20°C to freeze kill specimens prior to processing. Tubes and collection pots were labelled with the date of collection and a unique household number that also conferred geographic (district and village) information.

Collections were sorted via microscopy. Sand flies were separated from other collected arthropods and enumerated by sex and physiological status. Female and male sand flies were placed in separate microcentrifuge tubes containing RNALater with a maximum of 25 sand flies per tube. Specimens were stored in cryoboxes at -20°C.

Molecular analyses

DNA extraction. DNA was extracted from all female sand flies using the Qiagen DNeasy Blood and Tissue Kit per manufacturer's protocol. DNA samples were stored at -20°C. Molecular analyses were performed to confirm microscopic species identification of *P. argentipes*, for human DNA detection and *L. donovani* DNA detection in *P. argentipes* females.

Species identification via PCR. All female sand flies were identified using a PCR-RFLP protocol targeting the 18S rRNA coding genes for phlebotomine sandflies [24], which was adapted as described below. To differentiate *P. argentipes* from *P. papatasi*, and members of the *Sergentomyia babu* complex, PCR products underwent restriction enzyme digestion using *HinfI* and *HpalI* separately for one hour at 37°C. The total volume per reaction was 25µl, which include 12.5µl of Taq PCR master mix (Qiagen), 1.25 µl for each 10pmol/ µl primer (Forward 18S primer: 5'-TAGTGAAACCGCAAAAGGCTCAG-3'; Reverse 18S primer: 5'-CTCGGATGTGAGTCCTGTATT GT-3') and 10 µl of DNA sample. Digested products were run on a 2% agarose gel followed by ethidium bromide staining to detect bands of appropriate sizes corresponding to the three sandfly species.

Human DNA analysis. Female sand flies identified as *P. argentipes* by PCR-RFLP were analysed for the presence of human DNA using a qPCR protocol with primers targeting the cytochrome b (cytb) gene as described previously [25]. Quantitative detection of human DNA was performed on an Applied Biosystems 7500 Fast Real-Time PCR System according to the KAPA SYBR FAST Universal Master Mix recommended protocol. For standard curves,

human DNA obtained from purchased donor blood (Cambridge Bioscience) was serially diluted to provide a range of 1-0.0001 mg/µl. DNA free water was used as No Template Control (NTC) in each assay. A total of 5µl of DNA was used from each female *P. argentipes*. Samples were considered positive if Ct values were lower than the lower limit of detection of the assay (Ct< 30). All samples and controls were run in duplicate.

Leishmania donovani DNA analysis. Female sand flies identified as *P. argentipes* by PCR-RFLP were analysed for the presence of *L. donovani* DNA using a qPCR protocol with Taqman primers and probes as previously described and validated [26,27]. Quantitative detection of *Leishmania* DNA was performed on an Applied Biosystems 7500 Fast Real-Time PCR System using the conditions recommended by the KAPA Probe FAST Universal kit protocol. For standard curves, *L. donovani* DNA from the reference strain DD8 (MHOM/IN/80/DD8) was serially diluted to provide a range of $1-10^{-5}$ ng/µl. DNA from non-infected and experimentally infected sand flies from the LSHTM colony was used as negative and positive control, respectively. DNA free water was used as NTC in each assay. A total of 10µl of DNA was used from each female *P. argentipes*. Samples were considered positive if Ct values were lower than the limit of detection of the assay (Ct< 31). All samples and controls were run in duplicate.

Simulation of *Leishmania donovani* **RNA degradation.** In order to simulate the conditions that infective sand flies may experience during collections, sand flies were experimentally infected with *L. donovani* at high and low infection intensities of $3x10^6$ and $1x10^5$ amastigotes/ml heat-inactivated human blood, respectively, through a chick skin (equivalent to 3000 or 100 amastigotes/sand fly). Fully engorged females were maintained at 28° C, 80% relative humidity on 10% (w/v) sucrose for 8 days to allow the parasites to colonise the anterior midgut and produce metacyclic promastigotes, the infectious forms, confirmed by haemocytometer counting and morphology of parasites in Giemsa-stained smears of midgut homogenates (high dose infection (100% infection rate) $3.66 \times 10^4 \pm 2.00 \times 10^4$ average \pm standard deviation of parasites per fly; $52 \pm 20\%$ metacyclics per fly versus low dose infection: (70% infection rate) $4.92 \times 10^3 \pm 5.62 \times 10^4$ parasites; $5 \pm 10\%$ metacyclics per fly). Sand flies were then killed by brief exposure to chloroform vapour and kept under conditions they could experience after aspiration or collection by a CDC-LT until their arrival at the laboratory for RNA extraction and Real-Time-PCR analysis using *Leishmania*-specific ssrRNA primers (Table 1).

In the field, sand flies were placed in a cool bag (4°C) after collection at ambient temperatures of up to 40°C and then transported to the laboratory on a journey that could last up to 7 hours. In a CDC-LT, sand flies could be exposed to 40°C for up to 16 hours before being transferred to 4°C.

Leishmania donovani DD8 parasites were quantified from individual sand flies using a SYBR Green-based RT-qPCR protocol with Taqman primers and probes as previously described and validated [26,27]. Quantitative detection of *Leishmania* single subunit ribosomal RNA (*ssrRNA*) was performed on a Bio-Rad CFX Connect Real-Time PCR System. Briefly, RNA was recovered using a RNeasy Mini Kit (Qiagen) according to the manufacturer's instructions and stored at 80°C. Primers were designed using OligoPerfect Designer (Invitrogen Corp. Carlsbad, CA); the *ssrRNA* primer sequences and melting temperature (Tm) are as

Table 1. Storage conditions of experimentally infected sand flies to simulate field conditions following PKP aspiration or CDC-LT collection.

RNA extraction points	РКР	CDC
ГО	Processed immediately	Processed immediately
۲1	30 min, 40°C	16 hours, 40°C
Γ2	30 min, 40°C + 3 hours, 4°C	16 hours, 40° C + 3 hours, 4° C
Г3	30 min, 40°C + 7 hours, 4°C	16 hours, 40°C + 7 hours, 4°C

https://doi.org/10.1371/journal.pntd.0011200.t001

follows: Forward: CCATGTCGGATTTGGT; Reverse: CGAAACGGTAGCCTAGAG; melting temperature (Tm): 80.8°C. RNAs were reverse transcribed into cDNA using random hexamers (Roche Diagnostics) and Moloney Murine Leukaemia Virus Reverse Transcriptase (Invitrogen, Life Technologies) and quantified on a NanoDrop spectrophotometer. A SYBR Greenbased real-time PCR assay (QuantiTech SYBR Green Kit, Qiagen) was performed on white 96-well plates using a BioRad CFX Connect Real-Time System. 1 µl (100 ng) of sample cDNA was added to 9 µl of a master mix containing 5 µl of QuantiTect SYBR Green and 4 µl of nuclease-free water with primers (Guaranteed Oligos, Sigma-Aldrich) at a final concentration of 0.5 μM. Activation of the Qiagen Thermophilus aquaticus polymerase was performed at 95°C for 15 min. The PCR program included 44 cycles of denaturation at 95°C for 10 s, annealing at 54°C for 25 s and extension at 72°C for 30 s. SYBR Green fluorescent emission was measured at the end of the elongation step. Subsequently, a melting curve program was applied with a continuous fluorescent measurement starting at 70°C and ending at 95°C (ramping rate of 0.1° C s⁻¹). Crossing point values (Cp) were determined by the second derivative maximum method of the Bio-Rad CFX Maestro Software. cDNA from 10⁷ cultured Log phase L. donovani promastigotes was used a positive control and DNA free water was used as NTC in each assay.

Standard curves for *ssrRNA* transcripts were prepared using serial 10-fold dilutions of cultured *L. donovani* promastigotes $(10^7 \text{ to } 10^1)$ combined with an uninfected sand fly midgut. A linear regression for the standard curve was determined using the number of *Leishmania* parasites against the Ct values of *ssrRNA*. Subsequently, the fitted standard curves were used to estimate overall number of parasites in the sample.

Operational considerations

Time burden of sampling. To assess logistical constraints when using CDC-LT, MVA or PKP in a field setting, 24 additional trap-nights were attempted (8 per method) in a supplement study performed in the Saran villages in February 2019. The total time burden required to carry out collections for each method was recorded to the nearest minute for each household. For CDC-LTs, this was measured as the installation time burden. Additionally, the total time required to sort the samples in each collection pot according to the protocols implemented during the trapping comparison study was recorded to the nearest second.

Data collection and analysis

Data for each collection event (date of collection, district, village, sex and physiological state of sand flies) were recorded in a Microsoft Excel database. Descriptive statistics included analysis of total sand fly and total female mosquito data by collection method, district and village. Differences between the collection methods were investigated with respect to two primary outcomes—the total number of female sandflies and the number of female *P. argentipes* caught per collection—modelled as counts with a negative binomial likelihood, using mixed-effects regression. Independent and identically distributed random effects were used to account for clustering induced by multiple collections per household. Significance of pair-wise differences between methods was assessed at the 1% level using a Z-test with the Tukey adjustment for multiple comparisons. The number of male sand flies and number of female mosquitoes were also compared between methods as supplementary outcomes.

Data for the fourth round of collections were excluded from statistical analyses since specimens were lost during storage and species data were not obtainable. Descriptive statistical analyses were performed using STATA (version 17) and regression analyses using R (version 3.6.3 (2020-02-29)).

Results

CMC descriptive analysis of global data set

Collection site. Of a total of 576 individual collection events attempting to collect sand flies, 562 (97.7%) were completed successfully. Battery failure (for CDC-LTs left overnight) and inability to enter a household were the main reasons for unsuccessful collection events. Failures occurred across the four study villages, with one household having two failed collection events. A total of 6,809 sand flies were collected over the study period. At least one sand fly was collected from each household with an outlier of 396 total sand flies reported from one household in Ruchunpura (Nalanda) over the study period. Over 82% of collection events yielded at least one sand fly (465/562).

Fig 2 shows the number of specimens by collection site (Fig 2). Almost half of the sand flies collected were female (49.0%, n = 3,339). Of 3,232 female sand flies identified to species by PCR, 1,934 (59.8%) were *P. argentipes. Sergentomyia babu* accounted for 37.0% (n = 1,195) of females, and *P. papatasi* accounted for 3.2% (n = 103). The remaining 107 samples were unable to be identified to species. Although the proportion of total female sand flies collected in the two districts was roughly equal, the species composition differed. More *P. argentipes* females (n = 1,024, 52.9% versus n = 910, 47.1%) and *P. papatasi* females (n = 102, 99.0% versus n = 1, 1.0%) were collected in Saran than in Nalanda, respectively. Most of the *P. papatasi* females were collected from Rampur Jagdish (n = 95, 92.2%). Of 21,735 mosquitoes collected, 8,174 (37.6%) were female. Conversely, more mosquito females were collected in Nalanda (n = 5,728, 70.1%) than in Saran (n = 2,446, 29.9%).

Collection method. The total numbers of specimens by collection method for all four rounds is given in <u>S1 Table</u>. Fig 3 shows the mean number of sandflies collected by district,



Fig 2. Specimens collected by sex and collection location over the study period. (A) displays sand fly data, (B) displays mosquito data.

https://doi.org/10.1371/journal.pntd.0011200.g002



Fig 3. Mean female sand flies collected by method and village. Error bars indicate the 95% confidence interval of the mean.

https://doi.org/10.1371/journal.pntd.0011200.g003

village and by collection method. CDC-LTs collected the highest mean number of sandflies compared to MVA and PKP in both districts. CDC-LTs accounted for 57.5% of total sand flies collected, followed by PKP and MVA with 22.4% and 20.1%, respectively.

Fig 4 shows the mean number of female mosquitoes collected by district, village and by collection method (Fig 4). Bishambharpur had the lowest mean number of mosquitoes for all three collection methods. Collections in Dharampur using CDC-LTs had the highest overall mean female mosquitoes by village and method (32.5). The overall mean number of female mosquitoes per collection event was 14.5; CDC-LT yielded the greatest mean per collection event (16.96) while PKP yielded the lowest (10.0).

Further descriptive and regression analyses focused on *P. argentipes* females in the sand fly collections and were limited to the first three collection rounds due to post-collection loss of sand fly specimens from round four.

Statistical analyses for rounds 1-3 of CMC

The total numbers of specimens by round and village, and by collection method, for the first three rounds of collections are given in S_2 and S_3 Tables, respectively. Sampling round 2





https://doi.org/10.1371/journal.pntd.0011200.g004

yielded the highest mean for total sand flies, male sand flies, and female sand flies, with mean collections double those of round 1. Round 3 yielded the highest mean *P. argentipes* (Fig 5), with almost 2.5 times the collection of round 1, but the total mean sand flies collected was lower in round 3 than for round 2 for both males and females.

Negative binomial regression analysis comparing count of female sand flies and female *P. argentipes* showed CDC-LTs were significantly different from both MVA and PKP with increased rate of collection from CDC-LT collection events (both p<0.0001; Table 2). Of 511 collection events (excluding missing data), 75.6% (124/164) CDC-LTs collections resulted in the capture of at least one female *P. argentipes* compared to 43.2% (76/176) for MVA and 56.2% for PKP (87/171). Fig 6 shows the density of female *P. argentipes* by collection method. CDC-LTs captured the highest density of female *P. argentipes* per trap-night compared with MVA or PKP. However, the greatest number of female *P. argentipes* from a single collection event (n = 63) was collected by PKP. After adjusting for household clustering, CDC-LTs collected 3.41 times more female sand flies, 4.04 times more *P. argentipes*, and 8.23 more male sand flies than MVA (pairwise p<0.0001 for each). CDC-LTs also collected 3.08 times more female sand flies, 3.62 times more *P. argentipes*, and 7.21 more male sand flies than PKP



Fig 5. Mean sand flies collected per trap-night by collection round for the first three collection rounds (with 95% CI).

https://doi.org/10.1371/journal.pntd.0011200.g005

(pairwise p < 0.0001 for each) (Table 2). MVA collected significantly more female mosquitoes than PKP (pairwise p = 0.005), but there were no significant differences between CDC-LTs and PKP or MVA.

Molecular analyses

Detection of human and *Leishmania donovani* **DNA.** Of 1,934 female *P. argentipes*, 1,932 were analysed for the presence of a human DNA derived from a bloodmeal, and 156 (8.1%) were positive. Almost 75% of *P. argentipes* with human DNA were collected from the two Saran villages. As a proportion of total female *P. argentipes* caught, MVA collected significantly more female sand flies with human DNA compared with CDC-LTs and PKP (12.97% versus 5.65% and 7.40%, respectively, p-value<0.003).

Of 1,934 *P. argentipes* analysed for the presence of *L. donovani* DNA, none were positive. **Simulation of** *Leishmania donovani* **RNA degradation**. Processing infected sand flies immediately after killing them (condition 'T0') confirmed that the RT-PCR could detect high and low infection intensities, such that when the infection intensity was low there was an increase in the number of cycles for amplification (Fig 7A, 7B and 7C). However, even with low infection rates, most samples were amplified before 30 cycles indicating good sensitivity. For the 'worst case scenario', simulating CDC-LT samples where sand flies were collected overnight for 16h at 40°C and left in a cool bag for 7 hours (condition 'T3'), *L. donovani* RNA was detected in sand flies with either high or low infection intensities (Fig 7D and 7E).

		N	IRR (95% CI)	Z	P-value	
Total Female Sand Flies						
	MVA	685	0.293 (0.215-0.401)	-7.70	<0.0001	
	РКР	745	0.325 (0.238-0.442)	-7.13	<0.0001	
	CDC	1441	1	-	-	
P. argentipes Females						
	MVA	426	0.247 (0.161-0.380)	-6.39	<0.0001	
	РКР	468	0.276 (0.181-0.421)	-6.04	<0.0001	
	CDC	726	1	-	-	
Male Sand Flies						
	MVA	371	0.121 (0.086-0.171)	-12.13	<0.0001	
	РКР	435	0.138 (0.099–0.194)	-11.46	<0.0001	
	CDC	2019	1	-	-	
Female Mosquitoes*						
	MVA	2097	1.22 (0.915–1.648)	1.37	0.172	
	РКР	1440	0.791 (0.595–1.053)	-1.60	0.109	
	CDC	1843	1	-	-	

Table 2. Number of female sand flies, male sand flies, *P. argentipes* females, and female mosquitoes by collection method including incidence rate ratio (IRR) and negative binomial regression analyses. P-values are from a Z-test of the IRR being different from 0.

*Note that the model fit to the number of female mosquitoes failed to converge.

https://doi.org/10.1371/journal.pntd.0011200.t002

RT-PCR of high or low dose infected sand flies returned larger numbers of parasites when subjected to simulated PKP aspiration collection conditions compared with CDC-LT collection conditions (average \pm standard deviation: high dose infection: 2,742 \pm 3,376 versus 445 \pm 918, p = 0.0956; low dose infection: 295 \pm 530 versus 45 \pm 127, p = 0.0781)

Operational considerations

Time burden of sampling. On average, CDC-LTs took less time to install and to collect (6.29 mins) than the time taken to perform aspirations when using either MVA or PKP by 10.14 and 12.85 mins, respectively. The time taken to sort samples was influenced by the amount of material and number of target and non-target specimens within the collections. Although the mean time taken to sort CDC-LT collections (12.50 mins) was longer than for collections obtained using MVA or PKP by 1.29 and 7.19 mins, respectively, they also contained the highest number of female sand flies (n = 3), male sand flies (n = 6) and female mosquitoes (n = 157) than MVA (n = 0,1,92) or PKP (n = 1,0,3), respectively. Therefore, proportionally, the time invested in sorting compensated for the yield obtained.

Householder acceptance. Responses were received from all 48 householders who took part in the study. Amongst the respondents, 36 (75%) indicated they had a preference for CDC-LTs, 6 (13%) for MVA, 2 (4%) for PKP/MVA, and 4 (8%) had no preference. No householder indicated they had a preference solely for PKP collections. The most common reason given for a preference for CDC-LTs was the perception that fewer mosquitoes were present in their household following sampling. However, it is interesting to note that amongst householders who indicated a preference for either MVA or MVA/PKP, the most common explanation was also a perception that there were fewer mosquitoes present in the household after sampling. Three householders indicated that they felt CDC-LTs were the easiest collection technique, one considered it to be a less disruptive method, and two were unable to justify their choice. One respondent preferred MVA collections because they felt they had insufficient



Fig 6. Box plots showing the number of *P. argentipes* per collection event by method of collection. Medians indicated by horizontal line (at 0 for MVA and PKP) and outliers by black dots.

https://doi.org/10.1371/journal.pntd.0011200.g006

space in their home for a CDC-LT to be installed. When prompted for additional comments or complaints about the collection techniques, five householders responded that they found CDC-LTs to be too noisy at night and this caused disruption to their cattle. No further comments or complaints were received for either the PKP or MVA.

Discussion

Collection method comparison study

This is the first comprehensive field study that has made a direct comparison between CDC-LTs and two types of large battery-powered aspirators used in mosquito surveillance for the capture of *P. argentipes* females. The main finding that CDC-LTs collected an average of either 4.04 times or 3.62 times more *P. argentipes* females than either of the battery-powered aspirators, MVA or PKP, respectively, supports their continued use in entomological



Fig 7. RT-PCR amplification curves and corresponding parasite numbers for *Leishmania donovani* DD8 in sand flies 8 days post-infection with high (3000 amastigotes/sandfly) or low (<100 amastigotes/sandfly) doses of parasites. Flies were killed and stored under simulated conditions for Prokopack aspirator (PKP: orange) or CDC-LT collection (CDC: blue) and kept in a cool bag for 7 hours. (A, C, D) T0 amplification curves for infected sand flies (black), processed immediately after killing (A: High dose infection; C & D: Low dose infections), including a positive control of $1x10^6$ cultured Log phase *L. donovani* promastigotes (magenta) and a non-template negative control of DNA-free water (grey). (B, E) T3 amplification curves for high (B) and low (E) infected sand flies stored under 'worst case scenario' field conditions of 40°C for 30 min followed by 7 hours at 4°C (PKP), or 40°C for 16 hours followed by 7 hours at 4°C (CDC) before processing. (F, G) Parasite numbers per sand fly for T0 and T3 High (F) and low (G) dose infected sand flies, extrapolated from a standard curve. Intermediary storage conditions T1 and T2 (see Table 1) were tested and parasite numbers are included in F and G; the corresponding amplification curves are in S1 Fig. Data shown are representative of two independent experiments. Asterisks indicate values that are statistically significant (**P < 0.005; "P < 0.05; "P > 0.05) between the indicated groups using a two-sided unpaired Mann Whitney U t-test.

https://doi.org/10.1371/journal.pntd.0011200.g007

surveillance but other factors should be considered as described further below. Traditionally, CDC-LTs and/or mouth aspirators, rather than large battery-powered aspirators, are used in field studies performed in India to capture *P. argentipes* females [13,19,23,28–30]. However, a decision was made to avoid using mouth aspirators in the present study due to health and scientific concerns. Firstly, even if mouth aspirators are equipped with filters, the potential risk to the field team when making repeated aspirations over a long period of time, inevitably involving ingestion of small particles of organic debris while gently sucking sand flies from walls in a tube, was considered too high. Secondly, the degree of potential bias in efficiency between and amongst collectors, as well as problems relating to standardisation of the technique, was inappropriate for comparing collection methods [30]. Instead, two different battery-powered aspirators were compared using a standard operating procedure that was developed for the study.

As expected from previous work, capture rates of *P. argentipes* obtained during the present study were highly variable since they are known to be affected by several factors, including site, seasonality and climatic variables [13,30,31]. Such variability highlights the importance for testing different collection methods at the same sites and within the same timeframe, and for reducing sampling bias by using a Latin Square design as performed in the present study. Although superior for capturing *P. argentipes* and *Sergentomyia* females, CDC-LTs do not outperform the other collection methods for capturing *P. papatasi* or total mosquito females. It was not possible to make any species-specific comparisons between mosquito species because

mosquitoes were not identified in the present study. Collection methods for haematophagous insects are known to select for different behavioural traits, and species-specific differences between sand flies and mosquitoes in relation to phototropism (favouring capture by CDC-LTs), endophily and periodicity are not uncommon [32-35]. Since all collection methods took place indoors, each is likely to select females that are anthropophilic and endophilic, but collections may vary due to differences in periodicity and diurnal resting behaviour between species. In general, diurnal indoor resting behaviour of sand flies is understudied compared with mosquitoes but is also likely to be affected by a combination of factors such as microhabitat (temperature and humidity fluctuations that can occur diurnally, and vary according to house type) as well as selection pressure caused by long-term use of IRS [1,14,36,37]. A collection time window of 07.00 to 14.00 was selected in the present study, to be consistent with working hours of operational field teams, but it is possible that capture rates using either type of aspirator may have been higher if collections were made earlier in the day, 06.00–08.00, as typically performed in previous studies attempting to obtain higher numbers of P. argentipes females resting on walls [28,30,38]. Similarly, apart from Aedes mosquitoes that are active during the day, it is also likely that aspirations performed earlier in the day are likely to collect proportionally more *Culex* and *Anopheles* females [6,10,39,40].

The mean capture rates of *P. argentipes* females/trap night using CDC-LTs placed in human dwellings, of 4.2–4.8 or 6.0–6.1 *P. argentipes* females/trap night in Nalanda and Saran villages, respectively, were consistent with a mean capture rate of 5.5 *P. argentipes* females/trap night found in a previous study performed in Saran, but higher than the mean capture rate of 2.0 *P. argentipes* females/trap night reported in Muzaffarpur, Bihar, another VL endemic district bordering Saran [13,23] and in other sentinel sites in India where, even during the June-September peak, densities of *P. argentipes* females/trap night declined from 2014–2019 [41]. Although not included in the design of the present study, it is interesting to note that no impact of IRS could be indicated by comparing densities of *P. argentipes* females in CDC-LT collections in programmatic and non-programmatic villages in Bihar [16] so other entomological indicators, such as physiological status and/or infection rate may be more informative.

A complete data set, comprising speciation of *P. argentipes*, human DNA detection and *L. donovani* DNA detection, was obtained for three rounds (three 12x12 Latin squares) of collections (n = 144 collection events/method), rather than for all four rounds performed, and these data were used in statistical analyses. Significant differences between collection methods were detected for the primary outcome, number of *P. argentipes* females, since the study was still sufficiently powered to detect a mean difference of between 1.00–1.25 *P. argentipes* females using the data used to perform the sample size calculation [20,23]. It is interesting to note that, for the purposes of integrated vector-borne disease surveillance and multiplex pathogen screening, there were no significant differences between incidental captures of female mosquitoes using CDC-LTs compared with PKP or MVA.

To determine which human pathogens are circulating within an area, some MX studies use collection methods that specifically target blood-fed females to screen for infection. For studies measuring transmission, when it is important to maximise the likelihood of obtaining vectors infected with the human infectious stage of the parasite, then collection methods that target gravid vectors are used. Methods that target host-seeking females, such as CDC-LTs, tend to capture unfed females [32] which may be nulliparous (never blood-fed so will not be infected unless the pathogen is transovarially/transstadially transmitted) or parous (previously blood-fed, has laid eggs and seeking next bloodmeal so may also be infected/infectious). Some research groups include parity as an entomological outcome in their studies, but parity was not examined in the present study because it is time consuming, can be misclassified according to expertise, and is unlikely to be incorporated routinely in programmatic use.

Molecular analyses

Even though CDC-LTs collected proportionally less P. argentipes females with human DNA than MVA, there was no significant difference in the absolute number of *P. argentipes* females with human DNA captured. This finding is not unexpected since CDC-LT collections generally contain more unfed or gravid females than blood-fed sand flies [32], and it is consistent with a previous study performed in Bihar where proportionally less blood-fed P. argentipes females were captured by CDC-LTs compared with mouth aspirations [30]. In the present study, the mean rate of human DNA detected in P. argentipes females for all three collection methods was 8.1% with a mean rate of 5.4% for CDC traps placed inside houses. This rate is slightly higher than previously reported in Saran, where bloodmeals were detected in 3.2% and 2.6% of *P. argentipes* collected in CDC-LTs placed inside combined dwellings (where humans and cattle are in close proximity) and houses (with no cattle), respectively [13]. Previously, in Muzaffarpur, the mean percentage of blood-fed P. argentipes captured in CDC-LTs was 30.2% pre-intervention and 2.4% post-intervention with untreated bed nets [42]. Although it is possible that low feeding rates in Saran may be due to IRS intervention, there was no significant difference between the mean numbers of *P. argentipes* collected in households between Saran (VL endemic) and Nalanda (non-programmatic) villages, and the mean rate of human DNA detected in P. argentipes collected in Saran villages (11.4%) was significantly higher than for the non-endemic Nalanda villages (4.4%). There are several factors that may account for the differences, including: house construction (influencing sand fly entry and/or providing optimal conditions for resting), number of human inhabitants/house, type/number of livestock (either providing protection or increasing the risk of biting), and implementation of IRS (excito-repellency effect of synthetic pyrethroids may increase the risk of outdoor biting).

Human DNA detected in the sand flies could derive from a bloodmeal taken solely from a human host, or from a sand fly that had fed on multiple hosts since mixed bloodmeals have been reported for *P. argentipes* captured previously in Saran where human DNA was detected using cyt *b* PCR and reverse line blot in 164 of 288 (57%) blood-fed females [15]. The same study showed that the peak time to detect human DNA in *P. argentipes* in Saran villages was June-September which coincided with the sampling period performed in the present study [15]. In Muzaffarpur, the Human Bloodmeal Index was 81% pre-intervention and 19% post-intervention with untreated bed nets [42].

There are few studies measuring infection rates of *P. argentipes* with *L. donovani* DNA in India. Previously, CDC-LT collections performed in 2010 in Muzaffarpur showed that infection rates vary from 0.9% to 2.8% (12/422) according to the season [18]. In the present study, no *L. donovani* DNA was detected in each of the 1,934 sand flies individually tested from collections performed in 2018 (0/1024 from Saran and 0/910 from Nalanda). The inability to detect *L. donovani* DNA in *P. argentipes* in the Saran villages is surprising considering that VL and PKDL cases were reported on the KAMIS system in months leading up to and after when the work was performed. However, similar low infections rates in *P. argentipes* were also reported in eight VL sentinel sites undergoing routine entomological and IRS surveillance in three VL endemic States in India [41]. Of 14,775 *P. argentipes* females tested, *L. donovani* DNA was detected in only four females collected in the district of East Champaran, Bihar (3/ 991 and 1/704 females collected 2017 and 2018, respectively). No females were positive at any other of the sentinel sites suggesting that active transmission is low [41]. These data provide further evidence to support that efforts implemented during the accelerated kala-azar control programme may have successfully reduced the VL burden in endemic blocks.

It is important to note that differences in the sensitivity and specificity of the molecular assays used may affect differences in results. Many assays have been developed with different

molecular targets with a specificity variation between 29.6–100% and sensitivity between 91.3– 100% has been reported [27]. In addition, most assays used were developed to detect infections in human patients or dogs, rather than sand flies, and may cross-react with common insect protozoan parasites of the family Trypanosomatidae (e.g. *Leptomonas*). To avoid false positive or false negative results, it is important to use assays with targets specific for *L. donovani* and also use more than one target to confirm results, as performed in this study.

Future MX studies may require detection of *Leishmania* RNA in samples to quantify the burden of parasites, or assess their infectiousness with a metacyclic-specific target. Crucial to this is the assurance that parasite RNA in sand flies collected by PKP or CDC-LT will be able to robustly detect and quantify parasites. Here, we show that simulated PKP collection improves parasite quantitation in heavily infected flies, however, this degrades with transportation time such that by 7 hours at 4<u>°C</u> both collection methods performed equally. In flies infected with a low density of parasites, more akin to the dose naturally acquired by sand flies [43], parasite detection was also possible and their quantitation was consistent between PKP and CDC-LT collection conditions.

Operational considerations

Firstly, it should be noted that battery procurement and availability can be a limiting factor for all three methods used. It took a substantial effort to obtain the three different types of recommended batteries for the collection methods in India, but a supplier for large numbers of 6V 12amp batteries, as used for CDC-LTs, was later identified. Further evidence in favour for using CDC-LTs was the lower time burden for collecting specimens than either method of aspiration. This finding has some major implications for operational teams since it influences how many households can be sampled in one day: roughly 20 households may be sampled/day using CDC-LTs (including setting up and collecting traps) for every 9-10 or 10-11 aspirations using PKP or MVA, respectively. Sorting samples, and accurate recording of data, in the laboratory needs to be performed with precision and, although it took less time to sort samples collected using PKP than for the other methods of collection in our supplementary study, they contained less target material. It should be noted that the supplementary study performed to obtain time estimates took place in February, when sand fly numbers tend to be lower than June to September when the CMC study took place [13]. However, results of the CMC indicate that although collections using aspirators will contain less sand flies to process than collections using CDC-LTs, they will have similar numbers of female mosquitoes during the June to September peak and mosquitoes outnumber sand flies in the collections.

Importantly, more householders in the study expressed a preference for CDC-LTs than for the battery-powered aspirators in contrast to residents in Brazil who preferred MVA over CDC-LTs due to the nuisance of light and noise in the bedroom at night [10]. Continued consent of householders in India to allow indoor sampling using CDC-LTs in their homes should be monitored especially in light of some complaints relating to disturbance to animals expressed in the current study.

Recommendations

Based on the evidence obtained, CDC-LTs are the best method for collecting *P. argentipes* females in houses for transmission studies. However, capture rates are still low, and further research is required to increase collections of target species and to improve molecular methods. Specifically, in order to calculate the EIR of *P. argentipes*, a diagnostic test must be developed to identify the metacyclic stage of *L. donovani*. Although a laboratory protocol for detecting *L. mexicana* metacyclics in infected *Lutzomyia longipalpis* sand flies has been

published [44], it has not yet been validated in the field nor modified for other *Leishmania*/ sand fly species pairs.

Presently, molecular methods used for MX require skilled technical staff and a wellequipped laboratory and the cost of screening large numbers of vectors may be prohibitive for programmatic use thus reducing the role of MX for research purposes only [6]. To expand the flexibility of a MX system for programmatic use, and make it more cost-effective, integrating vector-borne disease surveillance should be considered. New multiplex methods for point-ofneed use are under development which make the possibility of integrating vector-borne disease surveillance a realistic and cost-effective proposition.

Supporting information

S1 Table. Specimens collected by collection method for all four rounds of the study. (DOCX)

S2 Table. Specimens collected by round and village. (DOCX)

S3 Table. Numbers and percentages of sand flies and mosquitoes collected by species, sex, and collection method during the first three collection rounds used in negative binomial regression analysis.

(DOCX)

S1 Fig. RT-PCR amplification curves for Leishmania donovani DD8 in sand flies 8 days post-infection with high (3000 amastigotes/sand fly) or low (<100 amastigotes/sand fly) doses of parasites. Flies were killed and stored under simulated conditions for Prokopack aspirator (PKP: orange) or CDC-LT collection (CDC: blue) and kept in a cool bag for up to 7 hours (Table 1). (A, B, C) T0 amplification curves for infected sand flies (black), processed immediately after killing (A: High dose infection; B & C: Low dose infections), including a positive control of 1x106 cultured Log phase L. donovani promastigotes (magenta) and a non-template negative control of DNA-free water (grey). (D, E) T1 amplification curves for high (D) and low (E) infected sand flies processed after exposure of killed flies to 40°C for 30 min (PKP) or 16 hours (CDC). (F, G) T2 amplification curves for high (F) and low (G) infected sand flies processed after exposure of killed flies to 40°C for 30 min and 4°C for 3 hours (PKP) or 40°C for 16 hours and 4°C for 3 hours (CDC). (H, I) T3 amplification curves for high (H) and low (I) infected sand flies killed and stored under 'worst case scenario' field conditions of 40°C for 30 min followed by 7 hours at 4°C (PKP), or 40°C for 16 hours followed by 7 hours at 4°C (CDC) before processing. (TIF)

Acknowledgments

The research team would like to thank the following individuals and organizations:

All householders in Nalanda and Saran who permitted access to their homes.

The Indian Council of Medical Research and the Government of India Health Ministry's Screening Committee for providing permission for the study. The National Centre for Vector Borne Disease Control (formerly the National Vector Borne Disease Control Programme) for their support and advice in placement of the research. Government of Bihar for providing population data of study villages.

Field workers and project administration staff who participated in this research.

The European Union's Horizon 2020 research and innovation programme, under grant agreement No 731060 (Infravec2), for provision of sand fly samples to optimise the PCR identification protocol.

Dr Tom Walker for advice on molecular tests.

Author Contributions

Conceptualization: Shannon McIntyre-Nolan, Pradeep Das, Mary M. Cameron.

Data curation: Miguella Mark-Carew.

Formal analysis: Miguella Mark-Carew, Emily S. Nightingale, Graham F. Medley.

Funding acquisition: Graham F. Medley, Mary M. Cameron.

- **Investigation:** Shannon McIntyre-Nolan, Kundan Kumar, Giorgia Dalla Libera Marchiori, Matthew E. Rogers, Mojca Kristan, Susana Campino.
- Methodology: Shannon McIntyre-Nolan, Emily S. Nightingale, Giorgia Dalla Libera Marchiori, Susana Campino.
- Project administration: Vijay Kumar, Mary M. Cameron.
- Resources: Vijay Kumar, Pradeep Das.
- Supervision: Matthew E. Rogers, Susana Campino, Graham F. Medley, Pradeep Das, Mary M. Cameron.
- Writing original draft: Shannon McIntyre-Nolan, Miguella Mark-Carew, Emily S. Nightingale, Susana Campino, Mary M. Cameron.
- Writing review & editing: Shannon McIntyre-Nolan, Vijay Kumar, Miguella Mark-Carew, Kundan Kumar, Emily S. Nightingale, Giorgia Dalla Libera Marchiori, Matthew E. Rogers, Mojca Kristan, Susana Campino, Graham F. Medley, Pradeep Das, Mary M. Cameron.

References

- 1. Chowdhury R, Kumar V, Mondal D, Das ML, Das P, Dash AP, et al. Implication of vector characteristics of Phlebotomus argentipes in the kala-azar elimination programme in the Indian sub-continent. Path Glob Health. 2016; 110(3):87–96. https://doi.org/10.1080/20477724.2016.1180775 PMID: 27376500
- World Health Organization. Regional Office for South-East Asia. Regional strategic framework for elimination of kala azar from the South-East Asia Region (2005–2015). New Delhi: WHO Regional Office for South-East Asia; 2005. Available from: https://apps.who.int/iris/handle/10665/205825
- 3. National Vector Borne Disease Control Programme. Accelerated plan for kala-azar elimination. Delhi, India: NVBDCP; 2017. Available from: https://nvbdcp.gov.in/WriteReadData/l892s/Accelerated-Plan-Kala-azar1-Feb2017.pdf
- World Health Organization. Global leishmaniasis surveillance, 2017–2018, and first report on 5 additional indicators. Weekly Epidemiological Record. 2020; 95:265–280. Available from: https://www.who.int/publications/i/item/who-wer9525
- World Health Organization. Report of Meeting of the Regional Technical Advisory Group (RTAG) on Visceral Leishmaniasis and the National Visceral Leishmaniasis Programme Managers of endemic Member States Regional Office for South-East Asia; 2020. Available from: <u>https://apps.who.int/iris/ handle/10665/340612</u>
- Cameron MM, Ramesh A. The use of molecular xenomonitoring for surveillance of mosquito-borne diseases. Philos Trans R Soc Lond B Biol Sci. 2021; 376(1818):20190816. <u>https://doi.org/10.1098/rstb.2019.0816</u> PMID: 33357052
- Ramzy RM, El Setouhy M, Helmy H, Farid HA, Gad AM, Weil GJ. Has Egypt Eliminated Lymphatic Filariasis. Am J Trop Med Hyg. 2007; 77(5):104–104. https://doi.org/10.4269/ajtmh.2007.77.5_Suppl. 0770000

- Pedersen EM, Stolk WA, Laney SJ, Michael E. The role of monitoring mosquito infection in the Global Programme to Eliminate Lymphatic Filariasis. Trends Parasitol. 2009; 25(7):319–27. https://doi.org/10. 1016/j.pt.2009.03.013 PMID: 19559649
- Subramanian S, Jambulingam P, Chu BK, Sadanandane C, Vasuki V, Srividya A, et al. Application of a household-based molecular xenomonitoring strategy to evaluate the lymphatic filariasis elimination program in Tamil Nadu, India. PLoS Negl Trop Dis. 2017; 11(4). <u>https://doi.org/10.1371/journal.pntd.</u> 0005519 PMID: 28406927
- Ramesh A, Cameron M, Spence K, Spaans RH, Melo-Santos MAV, Paiva MHS, et al. Development of an urban molecular xenomonitoring system for lymphatic filariasis in the Recife Metropolitan Region, Brazil. PLoS Negl Trop Dis. 2018; 12(10). https://doi.org/10.1371/journal.pntd.0006816 PMID: 30325933
- Ramirez AL, van den Hurk AF, Meyer DB, Ritchie SA. Searching for the proverbial needle in a haystack: advances in mosquito-borne arbovirus surveillance. Parasit Vectors. 2018; 11(1):320. <u>https://doi.org/10.1186/s13071-018-2901-x PMID: 29843778</u>
- Ready PD. Biology of phlebotomine sand flies as vectors of disease agents. Annu Rev Entomol. 2013; 58:227–50. https://doi.org/10.1146/annurev-ento-120811-153557 PMID: 23317043
- Poche D, Garlapati R, Ingenloff K, Remmers J, Poche R. Bionomics of phlebotomine sand flies from three villages in Bihar, India. J Vector Ecol. 2011; 36:S106–S17. <u>https://doi.org/10.1111/j.1948-7134.</u> 2011.00119.x PMID: 21366762
- Cameron MM, Acosta-Serrano A, Bern C, Boelaert M, den Boer M, Burza S, et al. Understanding the transmission dynamics of Leishmania donovani to provide robust evidence for interventions to eliminate visceral leishmaniasis in Bihar, India. Parasit Vectors. 2016;9. <u>https://doi.org/10.1186/s13071-016-1309-8 PMID: 26812963</u>
- Garlapati RB, Abbasi I, Warburg A, Poche DM, Poché RM. Identification of Bloodmeals in Wild Caught Blood Fed Phlebotomus argentipes (Diptera: Psychodidae) Using Cytochrome b PCR and Reverse Line Blotting in Bihar, India. J Med Entomol. 2012; 49(3):515–21. <u>https://doi.org/10.1603/me11115</u> PMID: 22679858
- Poché DM, Garlapati RB, Mukherjee S, Torres-Poche Z, Hasker E, Rahman T, et al. Bionomics of Phlebotomus argentipes in villages in Bihar, India with insights into efficacy of IRS-based control measures. PLoS Negl Trop Dis. 2018; 12(1):e0006168. https://doi.org/10.1371/journal.pntd.0006168 PMID: 29324760
- Tiwary P, Kumar D, Singh RP, Rai M, Sundar S. Prevalence of Sand Flies and Leishmania donovani Infection in a Natural Population of Female Phlebotomus argentipes in Bihar State, India. Vector Borne Zoonotic Dis. 2012; 12(6):467–72. https://doi.org/10.1089/vbz.2011.0808 PMID: 22217179
- Tiwary P, Kumar D, Mishra M, Singh RP, Rai M, Sundar S. Seasonal Variation in the Prevalence of Sand Flies Infected with Leishmania donovani. PLoS One. 2013; 8(4): e61370. <u>https://doi.org/10.1371/journal.pone.0061370 PMID: 23585896</u>
- Sardar AA, Chatterjee M, Jana K, Saha P, Maji AK, Guha SK, et al. Seasonal variation of sand fly populations in Kala-azar endemic areas of the Malda district, West Bengal, India. Acta Trop. 2020; 204:105358. https://doi.org/10.1016/j.actatropica.2020.105358 PMID: 31987778
- Mondragon-Shem K. Efficacy of CDC light and gravid traps for the collection of sandflies in Bhubaneswar, Orissa (India). M.Sc. Thesis, London School of Hygiene and Tropical Medicine. 2013.
- Vazquez-Prokopec GM, Galvin WA, Kelly R, Kitron U. A new, cost-effective, battery-powered aspirator for adult mosquito collections. J Med Entomol. 2009; 46(6):1256–9. https://doi.org/10.1603/033.046. 0602 PMID: 19960668
- 22. Torres MG, Weakley AM, Hibbert JD, Kirstein OD, Lanzaro GC, Lee Y. Ethanol as a potential mosquito sample storage medium for RNA preservation. F1000Res. 2019; 8:1431. <u>https://doi.org/10.12688/ f1000research.20162.1</u> PMID: 31497292
- Picado A, Das ML, Kumar V, Kesari S, Dinesh DS, Roy L, et al. Effect of Village-wide Use of Long-Lasting Insecticidal Nets on Visceral Leishmaniasis Vectors in India and Nepal: A Cluster Randomized Trial. PLoS Negl Trop Dis. 2010; 4(1). https://doi.org/e58710.1371/journal.pntd.0000587
- Tiwary P, Kumar D, Rai M, Sundar S. PCR-RFLP based method for molecular differentiation of sand fly species Phlebotomus argentipes, Phlebotomus papatasi, and Sergentomyia babu found in India. J Med Entomol. 2012; 49(6):1515–8. https://doi.org/10.1603/me12033 PMID: 23270185
- Sales KG, Costa PL, de Morais RC, Otranto D, Brandão-Filho SP, Cavalcanti Mde P, et al. Identification of phlebotomine sand fly blood meals by real-time PCR. Parasit Vectors. 2015; 8:230. https://doi.org/ 10.1186/s13071-015-0840-3 PMID: 25889289
- Vallur AC, Duthie MS, Reinhart C, Tutterrow Y, Hamano S, Bhaskar KR, et al. Biomarkers for intracellular pathogens: establishing tools as vaccine and therapeutic endpoints for visceral leishmaniasis. Clin Microbiol Infect. 2014; 20(6):O374–83. https://doi.org/10.1111/1469-0691.12421 PMID: 24237596

- Adams ER, Schoone G, Versteeg I, Gomez MA, Diro E, Mori Y, et al. Development and Evaluation of a Novel Loop-Mediated Isothermal Amplification Assay for Diagnosis of Cutaneous and Visceral Leishmaniasis. J Clin Microbiol. 2018;56(7). https://doi.org/10.1128/JCM.00386-18 PMID: 29695527
- Hati AK, Tandon N, Sinha A, Sur S, De N. A comparative field study of some sandfly sampling methods. Geneva: World Health Organization; 1987. Available from: https://apps.who.int/iris/handle/10665/ 59948
- 29. Hati AK, Sur S, De N, Dwivedi HN, World Health Organization Division of Vector Biology and Control. Occurrence of Phlebotomus argentipes at different heights above ground floor in cowsheds. Geneva: World Health Organization; 1989. Available from: https://apps.who.int/iris/handle/10665/59192
- Dinesh DS, Das P, Picado A, Davies C, Speybroeck N, Boelaert M, et al. The efficacy of indoor CDC light traps for collecting the sandfly Phlebotomus argentipes, vector of Leishmania donovani. Med Vet Entomol. 2008; 22(2):120–3. https://doi.org/10.1111/j.1365-2915.2008.00724.x PMID: 18498610
- Kesari S, Bhunia GS, Chatterjee N, Kumar VS, Mandal R, Das PK. Appraisal of Phlebotomus argentipes habitat suitability using a remotely sensed index in the kala-azar endemic focus of Bihar, India. Mem Inst Oswaldo Cruz. 2013;108 2:197–204. https://doi.org/10.1590/0074-0276108022013012 PMID: 23579800
- Alten B, Ozbel Y, Ergunay K, Kasap OE, Cull B, Antoniou M, et al. Sampling strategies for phlebotomine sand flies (Diptera: Psychodidae) in Europe. Bull Entomol Res. 2015; 105(6):664–78. https://doi.org/10. 1017/S0007485315000127 PMID: 26271257
- Hapairai LK, Plichart C, Naseri T, Silva U, Tesimale L, Pemita P, et al. Evaluation of traps and lures for mosquito vectors and xenomonitoring of Wuchereria bancrofti infection in a high prevalence Samoan Village. Parasit Vectors. 2015;8. https://doi.org/10.1186/s13071-015-0886-2 PMID: 26016830
- Pezzin A, Sy V, Puggioli A, Veronesi R, Carrieri M, Maccagnani B, et al. Comparative study on the effectiveness of different mosquito traps in arbovirus surveillance with a focus on WNV detection. Acta Trop. 2016; 153:93–100. https://doi.org/10.1016/j.actatropica.2015.10.002 PMID: 26466982
- World Health Organization. Operational manual on leishmaniasis vector control, surveillance, monitoring and evaluation. Geneva: World Health Organization; 2022. Available from: <u>https://apps.who.int/iris/ handle/10665/365615</u>
- Calderon-Anyosa R, Galvez-Petzoldt C, Garcia PJ, Carcamo CP. Housing Characteristics and Leishmaniasis: A Systematic Review. Am J Trop Med Hyg. 2018; 99(6):1547–54. <u>https://doi.org/10.4269/</u> ajtmh.18-0037 PMID: 30382013
- Wijerathna T, Gunathilaka N. Diurnal adult resting sites and breeding habitats of phlebotomine sand flies in cutaneous leishmaniasis endemic areas of Kurunegala District, Sri Lanka. Parasit Vectors. 2020; 13(1):284. https://doi.org/10.1186/s13071-020-04154-7 PMID: 32503610
- Kaul SM, Khera AK, Bora D. A comparative study on the optimum time and method of collection of Phlebotomus argentipes and other sandflies. (Diptera: Psychodidae). J Commun Dis. 1994; 26(3):172–6.
- Machani MG, Ochomo E, Amimo F, Kosgei J, Munga S, Zhou G, et al. Resting behaviour of malaria vectors in highland and lowland sites of western Kenya: Implication on malaria vector control measures. PLoS One. 2020; 15(2):e0224718. https://doi.org/10.1371/journal.pone.0224718 PMID: 32097407
- Seang-arwut C, Hanboonsong Y, Muenworn V, Rocklöv J, Haque U, Ekalaksananan T, et al. Indoor resting behavior of Aedes aegypti (Diptera: Culicidae) in northeastern Thailand. Parasit Vectors. 2023; 16(1):127. https://doi.org/10.1186/s13071-023-05746-9 PMID: 37060087
- Deb R, Singh RP, Mishra PK, Hitchins L, Reid E, Barwa AM, et al. Impact of IRS: Four-years of entomological surveillance of the Indian Visceral Leishmaniases elimination programme. PLoS Negl Trop Dis. 2021; 15(8):e0009101. https://doi.org/10.1371/journal.pntd.0009101 PMID: 34370731
- Picado A, Kumar V, Das M, Burniston I, Roy L, Suman R, et al. Effect of untreated bed nets on bloodfed Phlebotomus argentipes in kala-azar endemic foci in Nepal and India. Mem Inst Oswaldo Cruz. 2009; 104(8):1183–6. https://doi.org/10.1590/s0074-02762009000800018 PMID: 20140382
- Serafim TD, Coutinho-Abreu IV, Oliveira F, Meneses C, Kamhawi S, Valenzuela JG. Sequential blood meals promote Leishmania replication and reverse metacyclogenesis augmenting vector infectivity. Nat Microbiol. 2018; 3(5):548–55. https://doi.org/10.1038/s41564-018-0125-7 PMID: 29556108
- 44. Giraud E, Martin O, Yakob L, Rogers M. Quantifying Leishmania Metacyclic Promastigotes from Individual Sandfly Bites Reveals the Efficiency of Vector Transmission. Commun Biol. 2019; 2(1):84. <u>https://doi.org/10.1038/s42003-019-0323-8 PMID: 30854476</u>